

Citation: Walz Y, Wegmann M, Dech S, Vounatsou P, Poda J-N, N'Goran EK, et al. (2015) Modeling and Validation of Environmental Suitability for Schistosomiasis Transmission Using Remote Sensing. PLoS Negl Trop Dis 9(11): e0004217. doi:10.1371/journal.pntd.0004217

Editor: Justin V. Remais, Emory University, UNITED STATES

Received: June 13, 2015

Accepted: October 15, 2015

Published: November 20, 2015

Copyright: © 2015 Walz et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Epidemiological data were provided by the open access GNTD (http://www. gntd.org/). Landsat data can be downloaded from http://glovis.usgs.gov/. SRTM data can be downloaded from http://www.cgiar-csi.org/data/srtm-90m-digital-elevation-database-v4-1. RapidEye data are freely accessible only for scientific purposes and can be downloaded by the public RapidEye Science Archive of DLR https://resa.blackbridge.com/. The scientific project ID of this study is 474.

Funding: YW received funding through a PhD scholarship from the "Equal Opportunities for Women

RESEARCH ARTICLE

Modeling and Validation of Environmental Suitability for Schistosomiasis Transmission Using Remote Sensing

Yvonne Walz^{1,2}*, Martin Wegmann¹, Stefan Dech^{1,3}, Penelope Vounatsou^{4,5}, Jean-Noël Poda⁶, Eliézer K. N'Goran^{7,8}, Jürg Utzinger^{4,5}, Giovanna Raso^{4,5}

 Department of Remote Sensing, Institute for Geography and Geology, University of Würzburg, Würzburg, Germany, 2 United Nations University–Institute for Environment and Human Security, Bonn, Germany,
German Remote Sensing Data Centre, German Aerospace Centre, Oberpfaffenhofen, Germany,
Department of Epidemiology and Public Health, Swiss Tropical and Public Health Institute, Basel, Switzerland, 5 University of Basel, Basel, Switzerland, 6 Institut de Recherche en Sciences de la Santé, Ouagadougou, Burkina Faso, 7 Unité de Formation et de Recherche Biosciences, Université Félix Houphouët-Boigny, Abidjan, Côte d'Ivoire, 8 Centre Suisse de Recherches Scientifiques en Côte d'Ivoire, Abidjan, Côte d'Ivoire

* yvonne.walz@uni-wuerzburg.de

Abstract

Background

Schistosomiasis is the most widespread water-based disease in sub-Saharan Africa. Transmission is governed by the spatial distribution of specific freshwater snails that act as intermediate hosts and human water contact patterns. Remote sensing data have been utilized for spatially explicit risk profiling of schistosomiasis. We investigated the potential of remote sensing to characterize habitat conditions of parasite and intermediate host snails and discuss the relevance for public health.

Methodology

We employed high-resolution remote sensing data, environmental field measurements, and ecological data to model environmental suitability for schistosomiasis-related parasite and snail species. The model was developed for Burkina Faso using a habitat suitability index (HSI). The plausibility of remote sensing habitat variables was validated using field measurements. The established model was transferred to different ecological settings in Côte d'Ivoire and validated against readily available survey data from school-aged children.

Principal Findings

Environmental suitability for schistosomiasis transmission was spatially delineated and quantified by seven habitat variables derived from remote sensing data. The strengths and weaknesses highlighted by the plausibility analysis showed that temporal dynamic water and vegetation measures were particularly useful to model parasite and snail habitat suitability, whereas the measurement of water surface temperature and topographic variables

in Research and Training" programme at the University of Würzburg. This publication was funded by the German Research Foundation (DFG) and the University of Wuerzburg in the funding programme Open Access Publishing. All other authors received no specific funding for this work. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

did not perform appropriately. The transferability of the model showed significant relations between the HSI and infection prevalence in study sites of Côte d'Ivoire.

Conclusions/Significance

A predictive map of environmental suitability for schistosomiasis transmission can support measures to gain and sustain control. This is particularly relevant as emphasis is shifting from morbidity control to interrupting transmission. Further validation of our mechanistic model needs to be complemented by field data of parasite- and snail-related fitness. Our model provides a useful tool to monitor the development of new hotspots of potential schistosomiasis transmission based on regularly updated remote sensing data.

Author Summary

Schistosomiasis is a parasitic worm infection that is widespread in sub-Saharan Africa where people get in contact with open freshwater bodies. For many years, the strategy to control schistosomiasis was to prevent morbidity through deworming of school-aged children. Recently, transmission control has gained interest, which requires information where and when exactly transmission occurs. We investigated the potential of high-resolution remote sensing data to delineate potential transmission sites of schistosomiasis. Additionally, we characterized the habitat suitability for parasites and snails that are implicated in the schistosomiasis life cycle. Based on environmental field measurements in Burkina Faso and ecological data from the literature, functions of relative suitability were derived to determine the ecological relationship between the environment and snail and parasite fitness. These functions were employed to model the habitat suitability by using remote sensing variables that are aggregated to a habitat suitability index. We found that temporal dynamic of water bodies is one of the most relevant variables. Less relevant were topographic drainage lines. Our model also revealed significant relations with disease prevalence in different ecological zones of Côte d'Ivoire, and thus provides a useful tool to monitor new hotspots of disease transmission based on regularly updated remote sensing data.

Introduction

Schistosomiasis is a neglected tropical disease caused by blood flukes of the genus *Schistosoma*. From a public health perspective, schistosomiasis is considered the most important waterbased disease. Indeed, data from mid-2003 suggest that 779 million people were at risk of schistosomiasis with more than 200 million people infected, predominantly in sub-Saharan Africa [1]. An infection with schistosomes depends on the spatial and temporal distribution of specific freshwater snails that act as intermediate hosts and are the prerequisite that a *Schistosoma* parasite reaches the development stage to infect humans. The global strategy endorsed by the World Health Organization to control schistosomiasis is the large-scale administration of the antischistosomal drug praziquantel to at-risk populations to prevent morbidity [2,3]. The sustainability of this control strategy has been challenged, as there is rapid re-infection after deworming [4-6]. In recent years, a shift occurred from morbidity control to transmission control and local elimination, and hence, there is a stronger focus on intermediate host snails and

transmission sites, along with primary prevention tailored to specific social-ecological systems [7-9].

The use of satellite remote sensing data and techniques for risk profiling of environmentrelated diseases, including schistosomiasis, has increased considerably over the past 30 years [10]. In general, spatial and temporal properties of various remote sensing data and variables are investigated in relation to specific diseases or disease agents, such as intermediate hosts, parasites, or vectors for selected geographic regions of the world with the overall aim to support disease control and prevention strategies [10-17]. The standard approach for schistosomiasis risk profiling using remote sensing data presented in the literature is to relate the measure of Schistosoma infection prevalence-predominantly surveyed and geolocated at schools-to remote sensing measurements to model and spatially predict the risk of infection [16,18–21]. However, from a conceptual point of view, the application of remote sensing data aims at characterizing environmental conditions of potential disease transmission sites, which are in many cases spatially disjunct from the school location where epidemiological surveys are usually being conducted [21]. Few studies used remote sensing data to specifically characterize biophysical features of habitats in relation to snail prevalence, such as climatic variables in sub-Saharan Africa [22-24], and flooding and vegetation coverage in the People's Republic of China [25-27].

Specific habitat requirements of intermediate host snails are governed by environmental factors [28,29]. Beside the fact that intermediate host snails need an aquatic environment, it has been shown that certain conditions, such as water temperature, water flow velocity, and vegetation coverage, affect snail metabolism and fitness with consequences on species presence or absence [30-35]. A literature review on abiotic factors in relation to snail or parasite fitness revealed conditions that determine the abiotic environmental niche of a species and thus the suitability of a habitat as potential disease transmission site [36,37]. The habitat suitability index (HSI) is a methodological approach that allows to model, integrate, and summarize modeled environmental preferences, limits of tolerance, and behaviors of organisms in general. This methodological procedure was originally developed by the United States Fish and Wildlife Service (USFWS) to estimate the capacity of a habitat to support a species and quantify effects of land management alternatives on species habitats [38]. Thus far, USFWS has developed an HSI in over 150 species-specific models with the objective to support informed decision-making with respect to land management and species conservation [39]. Additionally, the HSI has been rigorously investigated for marine species [40,41], wildlife [42-44], and plant species [45-47], however, not yet for environment-related diseases such as schistosomiasis. Only few HSI models have been validated in a comprehensive manner [48], perhaps explained by the unavailability of adequate data to support validation [49]. It has been argued that HSI models belong to the most influential management tools and provide unique platforms to further explore species-environment relationships [50]. The key contribution of HSI models lies in quantifying both the quality and quantity of available habitats for selected species and is useful for monitoring dynamic habitat conditions over time, if current environmental data provide this information [51].

We investigated the potential of high-resolution remote sensing data to model environmental suitability for the transmission of schistosomiasis by means of a deductive mechanistic HSI modeling approach. We discuss strengths and weaknesses of remote sensing data for modeling environmental suitability, placing emphasis on (i) the spatial delineation of environmental conditions where disease transmission can potentially occur; (ii) quantitative prediction of environmental suitability within potential transmission sites; and (iii) transferability of an established model to different ecosystems. To support sustainable interruption of schistosomiasis transmission, spatial predictions of environmental suitability based on remote sensing data are required to expand the modeling perspective from school-located assessments to transmission site characteristics.

Methods

Ethics Statement

We present a secondary analysis with data derived from previously published studies that had been approved by the respective institutional review boards and national ethics committees; study site Ouagadougou [52,53], study site Man [54,55], and study site Taabo [56,57]. The studies were conducted according to national and international guidelines. All data received for the current analyses were anonymized.

Geography, Epidemiology, and Disease Control in the Study Areas

The study areas comprise three sites: (i) a large area around Ouagadougou (32,826 km²), the capital of Burkina Faso with a small sub-site around the city Ziniaré (3,155 km²); (ii) an area in the Man region (4,381 km²), located in the western part of Côte d'Ivoire; and (iii) a site around Taabo (8,476 km²) in south-central Côte d'Ivoire. The three sites represent a transect of ecozones from dry savannah in the North to tropical rainforest in the South, including flat and mountainous regions (Fig 1). The characteristic diurnal climate shows an annual precipitation gradient increasing from North (Ouagadougou: approximately 790 mm/year) to South (Taabo: approximately 1,230 mm/year and Man: approximately 1,630 mm/year) with mean temperatures above 25°C in all sites [58]. In Burkina Faso, the river network is subdivided by dammed lakes that provide surface water for manifold usage, especially during the dry season when rivers dry out temporarily. These man-made lakes strongly modify the river hydraulic conditions, resulting in reduced water flow velocity. Additionally, ponds are formed where rainwater fills topographic depressions. These play a crucial role for pastoral life and the transmission of schistosomiasis, especially where they constitute the only major water source [59,60].

Schistosomiasis is endemic in all three sites with moderate to high transmission rates at the time the data were collected [61]. The spatial distribution of schistosomiasis shows a typical focal pattern with prevalences at the unit of the school/village ranging between 0% and 100% [62]. While *Schistosoma haematobium* is the predominant species in Ouagadougou and Taabo, *S. mansoni* predominates in the Man region [63]. The main intermediate host snails are *Bulinus truncatus*, *Bu. globosus*, *Bu. senegalensis*, and *Bu. forskalii* for *S. haematobium*, and *Biomphalaria pfeifferi* for *S. mansoni* [64,65]. Based on snail surveys carried out in the 1980s and 1990s in Burkina Faso [64], 41% of intermediate host snails were found in small reservoirs, 34% in rivers, 20% in temporary ponds, 3% in irrigation channels, and 2% in natural lakes. These observations reveal the importance of small reservoirs with respect to the distribution of schistosomiasis in Burkina Faso [66]. The typical sites for potential disease transmission in Côte d'Ivoire are dam lakes, natural ponds, irrigation canals, and river confluences or bulges, where flow velocities of the rivers are slow, coupled with intense human-water contact patterns [65].

The surveillance and control of schistosomiasis are subject to supervision by national health authorities with the consequence that national borders delineate the respective efforts and applied practices between countries. In Burkina Faso, with the assistance from the Schistosomiasis Control Initiative (SCI) [67], nation-wide control of the disease has been implemented by the following steps: (i) identifying the most heavily infected regions; (ii) training local health staff and teachers; (iii) providing health education to the local population; and (iv) large scale administration of praziquantel to school-aged children. During mass treatment campaigns conducted between 2004 and 2006, more than 6 million school-aged children were treated





doi:10.1371/journal.pntd.0004217.g001

[68]. In Côte d'Ivoire a national control program has been established in 1998, but, due to limited funding and more than a decade of civil unrest, mass drug administration only became feasible in the recent past [69].

Remote Sensing Data and Derivation of Environmental Metrics

High-resolution remote sensing data from RapidEye (spatial resolution: 6.5 m) [70] with cloud coverage below 5% were acquired for the study sites in Côte d'Ivoire for January 2011, and for the sub-site Ziniaré at the end of the dry season (February 2010) and the end of the rainy season (October 2010). Additionally, six Landsat Thematic Mapper (TM) (30 m) scenes (paths 194–196, rows 51–52) were acquired for the study site in Ouagadougou during the dry (January and February 2010) and at the end of the rainy season (November and December 2010) from the United States Geological Survey (USGS) Global Visualization Viewer (http://glovis.usgs.gov/). Pre-processing of RapidEye and Landsat data involved the orthorectification, atmospheric, and topographic correction using the atmospheric correction tool ATCOR [71] within the CATENA (*lat.* for "chain") pre-processing tool of the German Aerospace Centre (DLR)

[72]. Clouds have been masked based on thresholds in the visible blue and thermal infrared band of Landsat 5 TM data [73]. Geolocation accuracy of the pre-processed Landsat 5 TM data was within 30 m sub-pixel level in reference to RapidEye data. The Advanced Spaceborne Thermal Emission and Reflection Radiometer (ASTER) global digital elevation model (GDEM) (30 m) data were downloaded from the Japan Space Systems [74] and used as basis for topographic analyses given a vertical root mean square error (RMSE) of 8.68 m [75].

Specific environmental metrics were derived from these remote sensing data to build up our habitat suitability model. A water mask has been derived from RapidEye and Landsat 5 TM imagery by thresholding the normalized difference water index (NDWI) [76] with a cut-off value set to zero, above which the spectral index delineates water. However, in the Man study site, the very small ponds and river lines, which are partly covered by dense riparian vegetation together with the exposure of dark rocks, contributed to misclassification. Hence, in this study area, the water bodies were mapped using a hierarchical procedure, including (i) a supervised classification of water/non-water employing a random forest classification algorithm [77]; (ii) the exclusion of topographic landscape elements improbable for the establishment of water bodies (curvature, slope) calculated from the ASTER GDEM; and (iii) the manual refinement of the water mask by visual inspection of RapidEye data in reference to very high-resolution data with spatial resolution below 2 m available in Google Earth [78]. Water surface temperature was measured by the Landsat 5 TM thermal band with an emissivity threshold of 0.98 [79]. The vegetation measure was derived from the RapidEye- and Landsat-based normalized difference vegetation index (NDVI), which responds to the change in the amount of green biomass and chlorophyll content. Additionally, several environmental metrics were derived from the ASTER GDEM, namely sinks, slope and stream order: topographic sinks that refer to locations where all neighboring values were equal to or greater than the center pixel value in the digital elevation model (DEM). Slope of the terrain was calculated with inclination in degrees, and a stream network of topographic water drainage was derived as a result of flow direction and flow accumulation analysis. The resulting stream network has been ordered according to Strahler [80] based on an accumulation threshold of 18,000 cells.

Validation Data: Environmental Field Data and Parasitological Data

During field work in the study site in Ouagadougou in March 2011, seven habitat types were identified according to the biotope classification of intermediate host snails in Burkina Faso [64]. A field verification form, adjusted from the standard Food and Agriculture Organization (FAO) Land Cover Classification System (LCCS) [81], guided the local analysis concerning potential disease transmission sites. Most relevant habitat characteristics were recorded at typical schistosomiasis transmission sites and are listed in <u>Table 1</u> and illustrated in <u>Fig 2</u>. Water persistence, vegetation coverage, and water flow velocity were estimated at the respective sites. Water body temperature was measured using a hand-held digital thermometer at the outer boundary of the water body and revealed temperatures ranging between 26°C (minimum) and 36°C (maximum). Although this measurement is not representative for the entire water body, it indicates a range of water surface temperature in this region. The site-specific estimate of environmental suitability is based on knowledge of habitat preferences of *S. haematobium* and *S. mansoni* cercariae and *Bulinus* and *Biomphalaria* snails [21].

Parasitological data have been accessed and extracted for the study sites from the Global Neglected Tropical Disease (GNTD) database (<u>http://www.gntd.org</u>) [<u>63,82</u>] with sample sizes of 74 locations for Ouagadougou, 75 for Man, and 38 for Taabo. The GNTD database includes relevant information about the date of survey, description of the survey location and population, number of people examined, diagnostic approach, and prevalence of infection [<u>63</u>]. The

Table 1. Example of field observations at typical schistosomiasis transmission sites. The measurements and estimates in the field were taken in the study site Ziniaré in Burkina Faso during the dry season in March 2011. NA indicates that this measure was not applicable. The typical transmission sites are illustrated in Fig 2. The habitat variable vegetation coverage was assessed based on the standard FAO LCCS form, flow velocity was categorized based on visual inspection, water temperature was measured using a hand-held digital thermometer, and estimation of the overall suitability of the transmission site resulted from knowledge on habitat preferences of species from the literature.

	Permanent water body	Rice field	Irrigated crops	River water	Seasonal water body	Dry river bed	Dry sink
Vegetation coverage	NA	75%	60%	NA	30%	20%	20%
Flow velocity	Stagnant	Stagnant	Slow flowing	Stagnant	NA	(Fluent: erosive river bed)	NA
Water tem-perature	26.2°C	28.5°C	32°C	31°C	NA	NA	NA
Estimated suitability	High	High	Moderate	Moderate	Moderate	Low	Low

doi:10.1371/journal.pntd.0004217.t001

final sample of parasitological data used in this study is a result of a preparatory selection procedure that removed surveys conducted before 1980 to respect the maximal life span of the parasites (30 years) [83] and surveys based on diagnostic approaches with a low sensitivity (e.g., direct fecal smear for *S. mansoni* infection) from the original database, as described by Schur et al. [84]. The data were used to validate modeled environmental suitability in different ecological zones.

Development and Composition of Environmental Suitability Model

A mechanistic modeling approach according to Brooks [50] was used to investigate the potential of remote sensing data to assess environmental suitability for the transmission of schistosomiasis. The model was developed for the sub-site Ziniaré in Burkina Faso, where high resolution remote sensing data and environmental field data were available, and tested for its transferability to the study sites Ouagadougou, Man, and Taabo (Fig 1). Specific habitat variables were derived from environmental metrics and parameterized based on functions that describe the relative suitability of each single habitat variable (Table 2). The derivation of functions of relative suitability is depending on the source of information and data provided [38].



Irrigated crops

Dry river bed

Dry sink



Fig 2. Illustration of field observation at typical schistosomiasis transmission sites.

doi:10.1371/journal.pntd.0004217.g002

Habitat variables [RS data and metrics]	Function of relative suitability	References	
Habitat stability w (weeks) for development of S. <i>mansoni</i> (a) and S. <i>haematobium</i> (b) [RapidEye/Landsat multitemporal water mask]	(a) $f(w) = \begin{cases} 0 \text{ for } 0 < w \le 4 \\ 0.5w = 2.5 \text{ for } 5 < w < 7 \end{cases}$	[<u>30,33,88]</u>	
	$ (a) f(w) = \begin{cases} 0.5w - 2.5 \text{ for } 5 \le w \le 7 \\ 1 \text{ for } 8 \le w \le 52 \\ 0 \text{ for } 0 < w \le 3 \end{cases} $		
	(b) $f(w) = \begin{cases} 0.5w - 2 \text{ for } 4 \le w \le 6 \end{cases}$		
	1 for $7 \le w \le 52$		
Water temperature T (°C) for S. mansoni (a), S. haematobium (b),	(0 for $T < 16$	[<u>31,32,34,35,89</u>]	
thermal emissivity]	(a) $f(T) = \begin{cases} -0.003 * \left(\frac{268}{T - 14.2} - 335\right) \text{ for } 16 \le T \le 35 \end{cases}$		
	0 for 35 < T (0 for $T < 17$		
	(b) $f(T) = \begin{cases} -0.006 * \left(\frac{295}{T - 15.3} - 174\right) \text{ for } 17 \le T \le 33 \end{cases}$		
	0 for 33 < T 0 for T < 16		
	(c)* $f(T) = \begin{cases} -4.095 + 0.368 T - 0.007 T^2 \text{ for } 16 \le T \le 35 \end{cases}$	≤ 35	
	0 for 35 < T (0 for $T < 17$		
	(d) $f(T) = \begin{cases} -2.350 + 0.208 T - 0.004 T^2 \text{ for } 17 \le T \le 33 \end{cases}$		
	0 for 33 < T		
Water flow velocity calculated from S [ASTER GDEM slope in degrees]	$f(S) = \int -5714.3 S + 1 \text{for} 0 \le S \le 0.00014$	[33,83,86,88]	
	$(0) = \begin{pmatrix} -0.0029 \text{S} + 0.2 \text{for S} > 0.00014 \end{pmatrix}$		
Water depth calculated from distance to shore x (m) [boundary of	$(-0.0043x + 1 \text{ for } 0 \le x \le 210)$	[<u>90]</u>	
napiuEye/Lanusai mulliemporai waler maskj	$f(\mathbf{x}) = \begin{cases} -0.000056\mathbf{x} + 0.088 \text{ for } 210 > \mathbf{x} \le 2000 \end{cases}$		
	0 for x > 2000		
Vegetation coverage V [NDVI derived from RapidEye/Landsat data]	$\int 0 \text{ for } V < 0$	[<u>30]</u>	
	$f(V) = \begin{cases} 3.33 \ V \text{ for } 0 \ge V \le 0.3 \end{cases}$		
	1 for V > 0.3		
Stream order (stream) [Topographic drainage lines derived from ASTER GDEM]	$f(stream) = 0.143 stream for 0 \le stream \le 7$	<u>[87]</u>	
Sink depth z (m) [ASTER GDEM-based sink depth]	f(z) = 0.005z + 0.11 for $1 < z < 222$	New variable	

Table 2. Functions of relative suitability used for modelling environmental suitability based on remote sensing metrics.

The derivation of these functions is explained in detail in the Supporting Information.

* This formula was originally derived for *Biomphalaria glabrata* strains in the laboratory. The validity of the formula was also tested in African snail species *B. pfeifferi* and *B. alexandrina* and agreed very well with the original data [34].

doi:10.1371/journal.pntd.0004217.t002

The habitat stability is defined as length of water persistence in weeks and was derived from multitemporal RapidEye data from the dry and wet season in the year 2010. The water temperature variable expresses the length of the prepatent period of parasites defined as response function for *S. mansoni* and for *S. haematobium* [34,35]. Additionally, the relation between water temperature and snail mortality was investigated for snails collected in the field and observed under laboratory conditions. The water flow velocity variable represents the slope measurement of ASTER GDEM data calculated into flow velocity following the Manning's velocity equation [85] and reflects a critical threshold at 0.3 m/s [86]. The relative suitability of water depth is expressed by the proxy measurement of Euclidean distance from the shoreline, which was calculated from the polyline boundary of the water masks from the dry and wet season. The relative suitability based on vegetation coverage was measured within a 200 m buffer area along detected water sites to capture irrigated agricultural fields (= vegetation suitability) and for water sites that seasonally dried out (= dry season vegetation suitability), as measured in the RapidEye satellite imagery. The NDVI threshold of 0.3 to mark high suitability stems from measurements in dry season RapidEye imagery and corresponds to areas of irrigated agriculture. Stream order resulted in hierarchical levels ranging from order 1 to 7 from a linear interpolation according to a previous study [87]. Sink depth suitability resulted from linear interpolation between the minimum and maximum depth measured in the study area. More details on the derivation of functions of relative suitability used in this study are provided in the Supporting Information. Relative suitability of each habitat variable was scaled between 0 and 1 and was considered highest when most favorable conditions for the coexistence of snail and parasite were given.

In a next step, habitat variables were combined to an overall HSI using a multi-criteria decision analysis (MCDA) [91]. In this study, the aggregation of habitat variables to the HSI was conducted using an additive priority function, as follows:

$$HSI = \sum_{i=1}^{m} a_i f$$

where *HSI* stands for the habitat suitability index and refers to the global priority of environmental suitability, *m* indicates the number of habitat variables, a_i describes the weighting of habitat variable *i*, and f_i gives the relative suitability of the habitat variable *i* between 0 and 1. Due to the lack of appropriate reference data to calibrate the model, a_i was weighted for each habitat variable by 1/m. The HSI was calculated separately for spatially superimposing habitat variables to capture (i) the relevant suitability at water sites (HSI_{water}) by aggregating the variables habitat stability, water temperature suitability, water flow suitability, water depth suitability, and dry season vegetation suitability, and (ii) relevant topographic characteristics where water can potentially accumulate (HSI_{pot.water}) by aggregating vegetation suitability within the 200 m buffer zone, stream suitability, potential flow suitability in drainage lines, where no water was detected by remote sensing data and sink depth suitability (Fig.3). The overall HSI value for a study site resulted from the juxtaposition of the HSI_{water} and HSI_{pot.water} derived for the defined spatial units (Fig.3).

Model Validation

The plausibility of modeled environmental suitability based on remote sensing data was validated in reference to on-site measurements and estimations presented in <u>Table 1</u>. Selected test sites were classified into high (HSI>0.67), moderate (HSI 0.33–0.67) and low (HSI<0.33) environmental suitability. Areas of no environmental suitability for schistosomiasis transmission correspond to regions where water cannot accumulate.

Additionally, the modeled environmental suitability was related to schistosomiasis prevalence based on the assumption that high environmental suitability in the catchment area of the surveyed location would be plausible if the measured prevalence was high and *vice versa*. It was the aim to elaborate whether this hypothesis can be confirmed and whether this locally developed model can be transferred to different ecozones in the Man and Taabo areas of western and south-central parts of Côte d'Ivoire, respectively. The linkage between modeled environmental suitability and schistosomiasis was established by spatially extending the point measurement of infection prevalence by a circular buffer with a radius of 5 km [1,92,93]. Within this buffer region, mean values of habitat variables and the composite HSI were extracted for each surveyed catchment area. The statistical relation between the mean environmental



Fig 3. Overview of single habitat variable suitability and the overall HSI. This Figure represents the output of modeled environmental suitability for schistosomiasis transmission at the sub-site of Ziniaré in Burkina Faso for the year 2010, based on which the model was developed.

doi:10.1371/journal.pntd.0004217.g003

suitability and the measured prevalence was assessed by the Spearman rank correlation coefficient [94]. Given that field data of parasite- and snail-related fitness were not available for this study to directly validate the HSI, disease prevalence data were used instead to validate environmental suitability as prevalence documents the outcome of the disease transmission process in the environment.

Results

Remotely Sensed Environmental Suitability

The resulting suitability of each habitat variable for water sites (HSI_{water}), potential water sites (HSI_{pot.water}), and the resulting HSI for the model development site of Ziniaré are illustrated in Fig 3. At water sites (HSI_{water}), highest suitability was measured for permanent water bodies and moderate suitability for temporary ones, which dried out during the dry season. Water temperature suitability represented the mean temperature suitability for *S. haematobium* and *S. mansoni* worms and *Bulinus* and *Biomphalaria* snails calculated from the dry and wet season water surface temperature. Most water bodies showed generally moderate to high temperature suitability for parasites and snails, whereas the riparian regions along the water bodies resulted in moderate to low temperature suitability. The relative suitability of flow velocity appeared to be heterogeneous, ranging from very low to very high. Nonetheless, one would expect stagnant or very slow moving water due to its topographic constitution as dam lake. Relative suitability of dry season vegetation coverage highlights areas of very high suitability, where dense vegetation covers a ground that was flooded during the rainy season corresponding to irrigated agriculture in this region.

At potential water sites (HSI_{pot.water}), the relative suitability of riparian vegetation coverage highlighted areas of irrigated agriculture and dense vegetation coverage as highly suitable, whereas highest suitability was reached when dry season and wet season vegetation coverage were high and irrigation was possible throughout the year. Topographic drainage lines resulted in high suitability at their lowest hierarchical order at the inflow and outflow of the great dam lake in the center of the study site and were modified by the overlaying suitability of flow velocity derived from the topographic slope. The sink habitat variable resulted primarily in very low suitability due to its flat character, whereas deeper sinks were masked due to their coverage with water.

The HSI showed a general discrimination between moderate to high suitability in and around water bodies and low suitability at topographic sinks. Individual water sites performed with variable suitability depending on the location in or around the water site. The zoomed-in part of the image in Fig.3 shows that high suitability pertained to areas at densely vegetated sites in the buffer zone of the water site and in certain areas of the water body, where especially water flow velocity was low. Moderate suitability referred to the littoral zone of permanent water levels, low vegetation cover buffer zones, and certain sectors of the topographic drainage line, whereas potential flow velocity resulted in high suitability. Low suitability resulted predominantly from topographic sinks, low vegetated region of seasonal flooded land, and sparsely vegetated regions in the buffer zone around the water site. Especially densely vegetated zones around permanent water sites resulted in high environmental suitability.

Plausibility of Modeled Environmental Suitability

The plausibility of modeled environmental suitability in reference to field estimates (<u>Table 1</u>) are summarized and illustrated in <u>Table 3</u> for the Ziniaré site within Ouagadougou. The measured HSI of 0.6 at the permanent waterbody (<u>Table 3</u>, row 1) was composited by the suitability

Type of habitat	Result of habitat variables	HSI value	Estimated habitat suitability at location (see <u>Table 1</u>)
Permanent water body	1 (HS); 0.41 (TS); 0.20 (FS); 0.81 (DS)	0.60 (moderate)	High
River water	1 (HS); 0 (TS); 0.19 (FS); 1 (DS)	0.55 (moderate)	Moderate
Seasonal water body	0.50 (HS); 0 (TS); 0.20 (FS); 0.50 (DS); 0.42 (dVS)	0.32 (low)	Moderate
Rice field	0.97 (bVS)	0.97 (high)	High
Irrigated crops	0.64 (bVS)	0.64 (moderate)	Moderate
Dry river bed	0.55 (bVS)	0.55 (moderate)	Low
Dry sink	0.08 (SiS)	0.08 (low)	Low

Table 3. Result of modeled suitability of single habitat variables and the HSI in comparison to field-based habitat suitability estimates for water sites and potential water sites.

Abbreviations: bVS = mean vegetation suitability within 200 m buffer zone of water; DS = water depth suitability; dVS = dry season riparian vegetation suitability; FS = water flow suitability; HS = suitability of habitat stability; SIS = sink suitability; TS = water temperature suitability

doi:10.1371/journal.pntd.0004217.t003

based on habitat stability (= 1), temperature suitability (= 0.41) resulting from surface temperature of 30°C in the dry season and 35°C in the wet season, flow suitability (= 0.20), and water depth suitability (= 0.81). At this site, the estimated high suitability did not meet the modeled HSI. In contrast, the HSI of 0.55 at the river water (Table 3, row 2) resulted in conformity with the field estimated moderate environmental suitability. The permanent water coverage was correctly detected by remote sensing data. However, the modeled temperature and flow suitability resulted in very low environmental suitability, which does not correspond to the at-site field measurements of stagnant water during the dry season and water temperature of 31°C. At the seasonal waterbody (Table 3, row 3), the HSI was slightly below the field estimated habitat suitability. However, the habitat variables showed plausible suitability.

The HSI of potential water sites (Table 3, rows 4–7) showed that, except for the dry sink test site, the suitability of mean vegetation coverage within a 200 m buffer zone of water was relevant to determine the HSI. The modeled HSI values reflected well the high and moderate suitability estimates for rice and irrigated crop sites, respectively. However, the estimated suitability of rice and irrigated crop sites resulted mainly from the irrigation practice, which varies between a permanently flooded rice field and crops that were irrigated by temporary flooding. This measure was not captured by remote sensing data in this case. The low environmental suitability estimated for the dry sink corresponded well to the modeled HSI of 0.08. However, the dry river bed was only captured due to its position within a 200 m buffer zone of water and resulted in a HSI value of 0.55 as a consequence of vegetation coverage at the site of the river bed. The HSI measure at this site would be expected to result from stream and potential flow suitability. However, this dry river bed was not detected by topographic drainage lines.

Model Transferability

The environmental suitability was plausible, as evaluated for the model development site Ziniaré. The HSI model was then applied to remote sensing data covering the larger study site Ouagadougou and the two study sites in Côte d'Ivoire (Man and Taabo). The transferability of the model from Ziniaré to other study sites was evaluated by the statistical rank correlation analysis between the level of *Schistosoma* spp. prevalence and modeled environmental

Table 4. Spearman rank correlation coefficients of *Schistosoma* prevalence and modelled environmental suitability. For this analysis, environmental suitability was extracted as mean value from a 5 km buffer zone around the measured prevalence. The corresponding confidence intervals are given in brackets. The columns represent the study sites and the rows represent the habitat variables.

	Ouagadougou	Man	Taabo
Suitability of habitat stability (HS)	-0.14 (-0.36, 0.09)	0.50* (0.30, 0.65)	0.32 (-0.00, 0.58)
Water temperature suitability (TS)	-0.08 (-0.30, 0.15)	0.47* (0.27, 0.63)	-
Water flow suitability (FS)	0.04 (-0.19, 0.27)	0.42* (0.21, 0.59)	0.30 (-0.17, 0.57)
Water depth suitability (DS)	-0.23 (-0.44, -0.01)	0.54* (0.36, 0.68)	-0.15 (-0.45, 0.18)
Dry season riparian vegetation suitability (dVS)	0.02 (-0.21, 0.25)	-	-
Mean vegetation suitability within 200 m buffer zone of water (bVS)	0.22 (0.00, 0.43)	0.47* (0.27, 0.63)	0.05 (-0.27, 0.36)
Stream suitability (StS)	0.09 (-0.14, 0.32)	0.50* (0.30, 0.65)	0.25 (-0.08, 0.53)
Sink suitability (SiS)	-0.12 (-0.34, 0.11)	0.20 (-0.03, 0.41)	-0.10. (-0.41, 0.23)
Potential water flow suitability in streams (pFS)	-0.09 (-0.31, 0.14)	0.42* (0.21, 0.59)	0.30 (-0.02, 0.56)
HSI	-0.12 (-0.34, 0.11)	0.45* (0.25, 0.62)	0.57* (0.31, 0.75)

*p < 0.01

doi:10.1371/journal.pntd.0004217.t004

suitability (<u>Table 4</u>). This relationship was not significant in the study site Ouagadougou, which shows that in this region other factors besides environmental suitability modify and thus explain the spatial heterogeneous distribution of disease prevalence (Fig 4).

In contrast, modeled environmental suitability for the Man region resulted in low suitability derived in the mountainous region in the northern part of this site and high suitability in the lowland of the southern part (Fig 4). In the Man region, areas of high suitability for disease transmission were predominantly represented by the course of rivers. Water buffer zones were largely covered by forest and did not correspond to high environmental suitability due to irrigated agricultural areas as derived in the savannah region of Burkina Faso. Nevertheless, a significant rank correlation to prevalence of schistosomiasis was found for all habitat variables except for sink suitability and for the HSI (Table 4).

Environmental suitability for schistosomiasis transmission in Taabo showed the highest correspondence between the level of environmental suitability and the measured schistosomiasis prevalence with a significant Spearman rank correlation coefficient of 0.57 (<u>Table 4</u>). This positive relation is confirmed by the spatial distribution of high prevalence rates in close proximity to the Bandama River and Lake Taabo, whereas low prevalence rates were predominantly distributed further away from these hotspots of environmental suitability with few exceptions close to the river outflow of Lake Taabo (Fig 4).

Discussion

In the current study, we developed a new modeling approach, which is based on a multitude of remotely sensed environmental metrics to delineate potential schistosomiasis transmission sites, and allows quantification of suitability for disease-related parasite and snail species. The validation of habitat variables in relation to field measurements and observations showed partial agreement. This validation procedure aimed at identifying strengths and weaknesses of remote sensing environmental metrics to assess environmental suitability for potential schistosomiasis transmission. Lessons learned are offered for discussion.

Habitat Variables

The high spatial resolution of RapidEye and Landsat 5 TM sensor provided relevant environmental information to detect the small-scale heterogeneity of water bodies in the study sites,





Fig 4. Habitat suitability index (HSI) and epidemiologic measurements at school locations for the three study sites Ouagadougou (Burkina Faso), Man and Taabo (Côte d'Ivoire), West Africa. In Ouagadougou and Taabo, schistosomiasis is mainly caused by S. haematobium, whereas in Man S. mansoni is widespread.

doi:10.1371/journal.pntd.0004217.g004

except in the Man region, where 6.5 m was still too coarse as water bodies predominantly consisted of small reservoirs and rivers, which, additionally, are often covered by large trees. Water temperature is considered an important habitat variable from an ecological point of view, as temperature is critical for both snail and parasite development. However, remotely sensed water surface temperature provides only to some extent the water temperature that directly impacts snail and parasite development and reproduction. Especially within isolated and shallow water bodies, extreme temperature variations were detected by Fisher and Mustard [95]. The measurement deviation in the range of 10°C between field water temperature and remotely sensed surface temperature might be caused by the inappropriate spatial resolution of the remotely sensed temperature measurement at very small water sites, such as the permanent river water site. Here, the 120 m spatial resolution of the thermal Landsat 5 TM band did not capture the linear structure of the river with width not exceeding 10 m and resulted in mixed pixel information of water and land surface temperature. This results in incorrect surface temperature measurements due to the calibration of surface emissivity for water. At this scale, very high resolution thermal remote sensing data would be more useful to derive water temperature of rivers and streams as shown by Torgersen et al. [96] using air-borne sensors.

Most optical remote sensing data capture visible and near infrared spectral reflectance, and are therefore well designed to measure vegetation coverage. In this study, vegetation was reasonably detected by high-resolution RapidEye data in comparison to field-based estimates of vegetation coverage for rice fields and irrigated crops. It must be noted that vegetation plays a crucial role for characterizing potential schistosomiasis transmission sites also in the sub-merged area, because submerged vegetation influences dissolved oxygen content of the water body. As a result, snail activity and reproduction are influenced by submerged vegetation [<u>30</u>]. Hyperspectral remote sensing data have demonstrated the ability to detect submerged aquatic vegetation [<u>97–99</u>], but are not yet regularly available as well as for large spatial extents.

The topography derived from the ASTER GDEM provided information on the topographic structure below the detected water level in all water bodies except for the great dam lake in the study site Ziniaré. Therefore, the measurement of slope for water sites often did not correspond to the water surface as intended for this model. This was well documented by strong heterogeneities of the slope measure within a dam lake, which was evident to have a flat surface of stagnant or very slow moving water. Additionally, the temporal dynamic of flow velocity between dry season and wet season as seen for permanent river water sites could not be captured by the single acquisition of a DEM. Nonetheless, suitability of flow velocity showed reasonable predictions for the large dam lake in the Ziniaré study site, and hence, was considered a useful proxy to highlight very flat zones within topographic streams. For large rivers, Kiel et al. [100] successfully derived water flow velocity using Shuttle Radar Topography Mission (SRTM) data. Topographically derived streams were very often not superimposing with the course of the actual river bed visible in high-resolution remote sensing data, which was further documented at the dry river bed test site. However, topographic sinks documented very well the course of river beds and could successfully detect a field measured sink.

Modeling Environmental Suitability

Environmental suitability for schistosomiasis transmission was modeled based on single habitat variables, which were parameterized by theoretical functions of relative suitability. These functions were derived from different background information, which is demonstrated by the following examples: habitat suitability related to water temperature was derived from laboratory-based measurements [34,35], whereas stream order suitability was characterized from field-based surveys [101] or spatial analysis [87]. The parasite-related water temperature suitability function was directly provided in the literature, whereas snail-related water temperature suitability was interpolated from measurements of snail mortality at given temperatures obtained from the extent literature [34,35]. The thresholds that characterize the range and course of functions given in this study were either cited in the literature (e.g., habitat stabilityrelated suitability), measured within high-resolution remote sensing data (e.g., vegetation coverage), or directly estimated in the field (e.g., water depth). These mixed sources of information can impact the resulting suitability of a habitat variable, because the functions have to be fitted to the modeled location. Due to the lack of appropriate field data on prevalence or fitness of parasites and snails, the parameterization of these functions could not be verified. However, these functions reflect the up-to-date knowledge on the ecology of schistosomiasis parasites and intermediate host snails and provide the basis for location-specific validations. The strengths and weaknesses of selected functions of relative suitability used in this study are discussed in the Supporting Information together with their derivation.

Habitat variables were aggregated to a composite HSI and assessed in relation to reference test sites scored into the classes low, moderate, and high environmental suitability. The model composition was adapted to the prerequisite of schistosomiasis transmission at the study site Ziniaré, and provides a transparent basis to reproduce and adjust the model. The implausible HSI at the seasonal water in dam lake and the dried river bed could be explained by the inappropriate spatial resolution of remotely sensed water temperature and topography. However, a precise identification of key factors determining the suitability of any particular habitat is difficult due to complex interactions [88]. These interactions are considered to some extent through the aggregation of single habitat variables to the HSI. It must be noted that the modeled environmental suitability represents fundamental ecological niches of *S. haematobium* and *S. mansoni* and *Bulinus* and *Biomphalaria*, which do not necessarily imply that species are abundant at an appropriate site. Parasites and snails can still be absent from apparently suitable habitats, because isolation of individual habitats and (re-)invasion are dictated by chance combinations of factors that permit snail dissemination [88].

Model Transferability and Public Health Relevance

The transfer of the model from its development site Ziniaré to other selected study sites showed that the specific landscape configuration of dam lakes and irrigated agriculture was highly relevant for potential schistosomiasis transmission in Burkina Faso. However, this was not given in the two study sites of Côte d'Ivoire. In the Taabo setting there is a large man-made lake, but there are no irrigated agricultural sites in close proximity, whilst in the Man area hardly any small dam lakes could be detected, but instead many rivers with dense forests at their watersides. Hence, the 200 m buffer zone around water bodies in Côte d'Ivoire was not considered representative of high environmental suitability given by the index in most areas, which hampered the direct transferability of the model from one geographic setting to another. However, water and potential water accumulation were well represented in all three study sites providing a reasonable basis to monitor potential disease transmission sites in a spatially explicit manner.

The hypothesis that the level of measured prevalence corresponds to the level of environmental suitability for schistosomiasis transmission was confirmed in the Taabo and Man settings, but declined for the study site Ouagadougou. It is clear that suitable environmental conditions provide the prerequisite that transmission of the disease may occur. However, there are factors that influence the relation between environmental suitability and schistosomiasis prevalence, such as local disease intervention measures [5,6], economic development [102], individual disease susceptibility [83,103], and human behavior [104,105].

The spatially explicit information of potential disease transmission sites highlighted in the current paper could provide a new tool for transmission control by guiding public health intervention measures directly to the place where humans become infected. The quantitative information of environmental suitability can support the prioritization of areas for transmission prevention. On the other hand, this model can complement research activities that elaborate on existing differences in disease transmission, occurrence of reservoir hosts, or habitats of parasites and intermediate host snails [7–9].

Outlook

Remote sensing data have shown the potential to spatially delineate the prerequisite that schistosomiasis transmission may occur through detection of water bodies and sites of potential water accumulation. Additionally, potential transmission sites have been evaluated for their suitability with respect to habitat preferences of disease-related snail and parasite species by means of remote sensing. The model presented in this study can be readily reproduced and adjusted by other experts as new data and information arise. A comprehensive validation of this mechanistic model approach would consist of a direct linkage between derived environmental suitability and field data of parasite- and snail-related fitness.

A predictive map of environmental suitability will support prevention and control measures against schistosomiasis, in line with the recent shift from morbidity to transmission control [Z–9]. Despite numerous constraints and limitations of the underlying models of environmental suitability, simple and even untested HSI models were continuously used for different kinds of decision-making procedures [50]. These maps could be useful to target specific locations for initial epidemiologic surveillance or to focus on areas of high reinfection potential for more regular re-treatment [20,106]. The model could provide a useful tool to monitor the development of new hotspots of potential disease transmission, as updated remote sensing data could directly detect changing environmental conditions such as the emergence of new dam lakes in remote areas.

Supporting Information

S1 Text. Derivation of Functions of Relative Suitability. (DOCX)

S2 Text. Discussion of Functions of Relative Suitability. (DOCX)

S3 Text. Additional References of Supporting Information. (DOCX)

Acknowledgments

The *Schistosoma* infection prevalence data were derived from the open-access Global Neglected Tropical Disease (GNTD) database (<u>http://www.gntd.org</u>). We thank the German Aerospace Center for providing RapidEye data from the RapidEye science archive. We thank Dr. Julia Kloos for additional comments on the manuscript.

Author Contributions

Conceived and designed the experiments: YW MW SD PV JNP EKN JU GR. Performed the experiments: YW. Analyzed the data: MW GR JU. Wrote the paper: YW MW GR JU. Read and approved the manuscript prior to submission: YW MW SD PV JNP EKN JU GR. Wrote the first draft of the manuscript: YW. Revised the manuscript: MW GR JU. Participated in the collection of the data in Burkina Faso and Côte d'Ivoire: JNP EKN GR JU. Overall responsibility of the integrity of the GNTD: PV.

References

- Steinmann P, Keiser J, Bos R, Tanner M, Utzinger J (2006) Schistosomiasis and water resources development: systematic review, meta-analysis, and estimates of people at risk. Lancet Infect Dis 6: 411–425. PMID: <u>16790382</u>
- 2. WHO (2002) Prevention and control of schistosomiasis and soil-transmitted helminthiasis: a report of a WHO expert committee. WHO Tech Rep Ser 912: 1–57.
- Doenhoff MJ, Hagan P, Cioli D, Southgate V, Pica-Mattoccia L, et al. (2009) Praziquantel: its use in control of schistosomiasis in sub-Saharan Africa and current research needs. Parasitology 136: 1825–1835. doi: <u>10.1017/S0031182009000493</u> PMID: <u>19281637</u>
- Utzinger J, Bergquist R, Xiao SH, Singer BH, Tanner M (2003) Sustainable schistosomiasis control the way forward. Lancet 362: 1932–1934. PMID: <u>14667754</u>

- Clements ACA, Bosqué-Oliva E, Sacko M, Landouré A, Dembélé R, et al. (2009) A comparative study of the spatial distribution of schistosomiasis in Mali in 1984–1989 and 2004–2006. PLoS Negl Trop Dis 3: e431. doi: <u>10.1371/journal.pntd.0000431</u> PMID: <u>19415108</u>
- Zhang Z, Zhu R, Ward MP, Xu W, Zhang L, et al. (2012) Long-term impact of the World Bank Loan Project for schistosomiasis control: a comparison of the spatial distribution of schistosomiasis risk in China. PLoS Negl Trop Dis 6: e1620. doi: 10.1371/journal.pntd.0001620 PMID: 22530073
- Stothard JR, Chitsulo L, Kristensen TK, Utzinger J (2009) Control of schistosomiasis in sub-Saharan Africa: progress made, new opportunities and remaining challenges. Parasitology 136: 1665–1675. doi: 10.1017/S0031182009991272 PMID: 19814845
- Utzinger J, N'Goran EK, Caffrey CR, Keiser J (2011) From innovation to application: social-ecological context, diagnostics, drugs and integrated control of schistosomiasis. Acta Trop 120 (Suppl 1): S121–S137. doi: 10.1016/j.actatropica.2010.08.020 PMID: 20831855
- Rollinson D, Knopp S, Levitz S, Stothard JR, Tchuem Tchuenté L-A, et al. (2013) Time to set the agenda for schistosomiasis elimination. Acta Trop 128: 423–440. doi: <u>10.1016/j.actatropica.2012.04.</u> 013 PMID: <u>22580511</u>
- Kalluri S, Gilruth P, Rogers D, Szczur M (2007) Surveillance of arthropod vector-borne infectious diseases using remote sensing techniques: a review. PLoS Pathog 3: e116.
- 11. Hay SI, Packer MJ, Rogers DJ (1997) The impact of remote sensing on the study and control of invertebrate intermediate hosts and vectors for disease. Int J Remote Sens 18: 2899–2930.
- Beck LR, Lobitz BM, Wood BL (2000) Remote sensing and human health: new sensors and new opportunities. Emerg Infect Dis 6: 217–226. PMID: <u>10827111</u>
- **13.** Hay SI, Randolph SE, Rogers DJ (2000) Remote sensing and geographical information systems in epidemiology—guest editors´ preface. Adv Parasitol 47: xi–xii.
- 14. Brooker S (2002) Schistosomes, snails and satellites. Acta Trop 82: 207–214. PMID: 12020894
- 15. Yang GJ, Vounatsou P, Zhou XN, Utzinger J, Tanner M (2005) A review of geographic information system and remote sensing with applications to the epidemiology and control of schistosomiasis in China. Acta Trop 96: 117–129. PMID: <u>16112638</u>
- Simoonga C, Utzinger J, Brooker S, Vounatsou P, Appleton CC, et al. (2009) Remote sensing, geographical information system and spatial analysis for schistosomiasis epidemiology and ecology in Africa. Parasitology 136: 1683–1693. doi: 10.1017/S0031182009006222 PMID: 19627627
- 17. Tran A, Goutard F, Chamaillé L, Baghdadi N, Lo Seen D (2010) Remote sensing and avian influenza: a review of image processing methods for extracting key variables affecting avian influenza virus survival in water from Earth observation satellites. Int J Appl Earth Obs Geoinf 12: 1–8.
- Seto E, Xu B, Liang S, Gong P, Wu W, et al. (2002) The use of remote sensing for predictive modeling of schistosomiasis in China. Photogramm Eng Remote Sens 68: 167–174.
- Stensgaard AS, Jorgensen A, Kabatereine NB, Malone JB, Kristensen TK (2005) Modeling the distribution of *Schistosoma mansoni* and host snails in Uganda using satellite sensor data and geographical information systems. Parassitologia 47: 115–125. PMID: <u>16044680</u>
- Brooker S (2007) Spatial epidemiology of human schistosomiasis in Africa: risk models, transmission dynamics and control. Trans R Soc Trop Med Hyg 101: 1–8. PMID: <u>17055547</u>
- Walz Y, Wegmann M, Dech S, Raso G, Utzinger J (2015) Risk profiling of schistosomiasis using remote sensing: approaches, challenges and outlook. Parasit Vectors 8: 163. doi: <u>10.1186/s13071-015-0732-6</u> PMID: <u>25890278</u>
- Moodley I, Kleinschmidt I, Sharp B, Craig M, Appleton C (2003) Temperature-suitability maps for schistosomiasis in South Africa. Ann Trop Med Parasitol 97: 617–627. PMID: <u>14511560</u>
- 23. Stensgaard AS, Jorgensen A, Kabatereine NB, Rahbek C, Kristensen TK (2006) Modeling freshwater snail habitat suitability and areas of potential snail-borne disease transmission in Uganda. Geospat Health 1:93–104. PMID: <u>18686235</u>
- 24. Stensgaard AS, Utzinger J, Vounatsou P, Hürlimann E, Schur N, et al. (2013) Large-scale determinants of intestinal schistosomiasis and intermediate host snail distribution across Africa: does climate matter? Acta Trop 128: 378–390. doi: 10.1016/j.actatropica.2011.11.010 PMID: 22142789
- **25.** Guo JG, Vounatsou P, Cao CL, Utzinger J, Zhu HQ, et al. (2005) A geographic information and remote sensing based model for prediction of *Oncomelania hupensis* habitats in the Poyang Lake area, China. Acta Trop 96: 213–222. PMID: 16140246
- 26. Yang GJ, Vounatsou P, Tanner M, Zhou XN, Utzinger J (2006) Remote sensing for predicting potential habitats of *Oncomelania hupensis* in Hongze, Baima and Gaoyou lakes in Jiangsu province, China. Geospat Health 1: 85–92. PMID: <u>18686234</u>

- Zhang Z, Bergquist R, Chen D, Yao B, Wang Z, et al. (2013) Identification of parasite-host habitats in Anxiang county, Hunan province, China based on multi-temporal China-Brazil Earth Resources Satellite (CBERS) images. PLoS One 8: e69447. doi: <u>10.1371/journal.pone.0069447</u> PMID: <u>23922712</u>
- Poda JN, Sawadogo LL, Sellin B, Sanogo S (1996) Dynamique des populations de Bulinus truncatus rohlfsi Clessin, 1886, dans le barrage de Dyoro en zone nord-soudanienne du Burkina Faso. Rev Agronom Afr 8: 61–68. (in French).
- Poda JN, Sellin B, Sawadogo L (1994) Dynamique des populations de Bulinus senegalensis, MULLER, 1781, dans une mare temporaire de la zone nord-soudanienne du Burkina Faso. Rev Elev Méd Vét Pays Trop 47: 375–378. (in French).
- **30.** Abdel-Malek E (1958) Factors conditioning the habitat of bilharziasis intermediate hosts of the family *Planorbidae*. Bull World Health Organ 18: 785–818. PMID: <u>13573113</u>
- **31.** Shiff CJ (1964) Studies on *Bulinus* (Physopsis) *globosus* in Rhodesia. I The influence of temperature on the intrinsic rate of natural increase. Ann Trop Med Parasitol 58: 94–105. PMID: 14149705
- **32.** Shiff CJ, Garnett B (1967) The influence of temperature on the intrinsic rate of natural increase of the freshwater snail *B. pfeifferi*. Arch Hydrobiol 62: 429–438.
- Appleton CC (1978) Review of the literature on abiotic factors influencing the distribution and lifecycles of bilharziasis intermediate host snails. Malacol Rev 11: 1–25.
- Pflüger W (1980) Experimental epidemiology of schistosomiasis. I. The prepatent period and cercarial production of *Schistosoma mansoni* in *Biomphalaria* snails at various constant temperatures. Z Parasitenkd 63: 159–169. PMID: <u>7456642</u>
- **35.** Pflüger W, Roushdy MZ, El Emam M (1984) The prepatent period and cercarial production of *Schistosoma haematobium* in *Bulinus truncatus* (Egyptian field strains) at different constant temperatures. Z Parasitenkd 70: 95–103. PMID: <u>6695537</u>
- Malone JB (2005) Biology-based mapping of vector-borne parasites by geographic information systems and remote sensing. Parassitologia 47: 27–50. PMID: <u>16044674</u>
- 37. Kearney M (2006) Habitat, environment and niche: what are we modelling? Oikos 115: 186–191.
- **38.** USFWS (1981) Standards for the development of habitat suitability index models. Washington D.C.: United States Fish and Wildlife Service.
- **39.** USFWS (2014) USGS National Wetlands Research Center Digital Library: Habitat Suitability Index Models. Biological Report: U.S. Department of the Interior, U.S. Geological Survey.
- Brown SK, Buja KR, Jury SH, Monaco ME, Banner A (2000) Habitat suitability index models for eight fish and invertebrate species in Casco and Sheepscot Bays, Maine. North Am J Fish Manag 20: 408– 435.
- **41.** Vinagre C, Fonseca V, Cabral H, Costa MJ (2006) Habitat suitability index models for the juvenile soles, *Solea solea* and *Solea senegalensis*, in the Tagus estuary: defining variables for species management. Fish Res 82: 140–149.
- Thomasma LE, Drummer TD, Peterson RO (1991) Testing the habitat suitability index model for the fisher. Wildlife Soc B 19: 291–297.
- **43.** Mitchell MS, Zimmerman JW, Powell RA (2002) Test of a habitat suitability index for black bears in the southern Appalachians. Wildlife Soc B 30: 794–808.
- 44. Dussault C, Courtois R, Ouellet J-P (2006) A habitat suitability index model to assess moose habitat selection at multiple spatial scales. Can J For Res 36: 1097–1107.
- Store R, Kangas J (2001) Integrating spatial multi-criteria evaluation and expert knowledge for GISbased habitat suitability modelling. Landscape Urban Plan 55: 79–93.
- Rüger N, Schlüter M, Matthies M (2005) A fuzzy habitat suitability index for *Populus euphratica* in the northern Amudarya delta (Uzbekistan). Ecol Model 184: 313–328.
- Williams NSG, Hahs AK, Morgan JW (2008) A dispersal-constrained habitat suitability model for predicting invasion of alpine vegetation. Ecol Appl 18: 347–359. PMID: <u>18488601</u>
- Duncan BW, Breininger DR, Schmalzer PA, Larson VL (1995) Validating Florida scrub jay habitat suitability model, using demography data on Kennedy Space Center. Photogramm Eng Remote Sens 11: 1361–1370.
- Schamberger ML, O'Neil LJ (1986) Concepts and constraints of habitat-model testing. In: Verner J, Morrison ML, Ralph CJ, editors. Wildlife 2000: Modelling habitat relationships of terrestrial vertebrates. Wisconsin, USA: University of Wisconsin Press. pp. 5–10.
- 50. Brooks RP (1997) Improving habitat suitability index models. Wildl Soc Bull 25: 163–167.
- **51.** Ortigosa GR, De Leo GA, Gatto M (2000) VVF: integrating modelling and GIS in a software tool for habitat suitability assessment. Environ Model Softw 15: 1–12.

- 52. Brasseur P, Drhuile P (1984) [Geographic dispersion of bilharzial endemy in 3 departments of Burkina Faso (ex-Upper Volta)]. Bull Soc Pathol Exot Filiales 77: 673–677. (in French). PMID: 6543159
- Garba A, Campagne G, Poda JN, Parent G, Kambire R, et al. (1999) [Schistosomiasis in the region of Ziga (Burkina Faso) before the construction of a dam]. Bull Soc Pathol Exot 92: 195–197. (in French). PMID: <u>10472448</u>
- Utzinger J, N'Goran EK, Ossey YA, Booth M, Traore M, et al. (2000) Rapid screening for Schistosoma mansoni in western Côte d'Ivoire using a simple school questionnaire. Bull World Health Organ 78: 389–398. PMID: <u>10812739</u>
- 55. Raso G, Matthys B, N'Goran EK, Tanner M, Vounatsou P, et al. (2005) Spatial risk prediction and mapping of *Schistosoma mansoni* infections among schoolchildren living in western Côte d'Ivoire. Parasitology 131: 97–108. PMID: <u>16038401</u>
- N'Goran EK, Diabate S, Utzinger J, Sellin B (1997) Changes in human schistosomiasis levels after the construction of two large hydroelectric dams in central Côte d'Ivoire. Bull World Health Organ 75: 541–545. PMID: <u>9509626</u>
- N'Goran EK, Utzinger J, Traorée M, Lengeler C, Tanner M (1998) [Use of a questionnaire for quick identification of the principal foci of urinary bilharziasis in central Ivory Coast]. Med Trop (Mars) 58: 253–260. (in French).
- Hijmans RJ, Cameron SE, Parra JL, Jones PG, Jarvis A (2005) Very high resolution interpolated climate surfaces for global land areas. Int J Climatol 25: 1965–1978.
- 59. Dipama JM (2010) The biophysical environment: hydrology. In: Thiombiano A, Kampmann D, editors. Biodiversity Atlas of West Africa, Volume II: Burkina Faso. Ouagadougou & Frankfurt/Main.
- 60. Savane I (2010) The biophysical environment: water resources. In: Konaté S, Kampmann D, editors. Biodiversity Atlas of West Africa, Volume III: Côte d'Ivoire. Abidjan & Frankfurt/Main.
- **61.** WHO (2010) Schistosomiasis control at a glance. World Health Organization—Regional Office for Africa. pp. 2.
- Assaré RK, Lai Y-S, Yapi A, Tian-Bi YNT, Ouattara M, et al. (2015) The spatial distribution of Schistosoma mansoni infection in four regions of western Côte d'Ivoire. Geospat Health 10: 69–79.
- **63.** Hürlimann E, Schur N, Boutsika K, Stensgaard AS, Laserna de Himpsl M, et al. (2011) Toward an open-access global database for mapping, control, and surveillance of neglected tropical diseases. PLoS Negl Trop Dis 5: e1404. doi: <u>10.1371/journal.pntd.0001404</u> PMID: <u>22180793</u>
- 64. Poda JN (1996) Spatial distribution of intermediate hosts of schistosomes in Burkina Faso: factors influencing the dynamics of *Bulinus truncatus rohlfsi* and *Bulinus senegalensis* [PhD thesis]: University of Ouagadougou.
- 65. Kinanpara K, Yves BK, Félix KK, Edia EO, Théophile G, et al. (2013) Freshwater snail dynamics focused on potential risk of using urine as fertilizer in Katiola, an endemic area of schistosomiasis (Ivory Coast; West Africa). J Entomol Zool Stud 1: 110–115.
- 66. Boelee E, Cecchi P, Kone A (2009) Health impacts of small reservoirs in Burkina Faso. Colombo: International Water Management Institute. 50 p.
- Fenwick A, Webster JP, Bosque-Oliva E, Blair L, Fleming FM, et al. (2009) The Schistosomiasis Control Initiative (SCI): rationale, development and implementation from 2002–2008. Parasitology 136: 1719–1730. doi: 10.1017/S0031182009990400 PMID: 19631008
- MoH, PNLSc (2010) Plan d'action 2011 du programme national de lutte contre les schistosomoses. Ouagadougou: Ministère de la Santé Burkina Faso—Programme National de Lutte contre les Schistosomoses.
- Tchuem Tchuenté LA, N'Goran EK (2009) Schistosomiasis and soil-transmitted helminthiasis control in Cameroon and Côte d'Ivoire: implementing control on a limited budget. Parasitology 136: 1739– 1745. doi: <u>10.1017/S0031182009005988</u> PMID: <u>19490723</u>
- 70. Blackbridge (2015) Satellite Imagery Product Specifications. Berlin.
- 71. Richter R, Schläpfer D (2012) Atmospheric / topographic correction for airborne imagery. Wessling: German Aerospace Center (DLR).
- 72. DLR (2014) CATENA. Wessling: German Aerospace Center (DLR).
- 73. Martinuzzi S, Gould WA, Ramos Gonzalez OM (2007) Creating cloud-free Landsat ETM+ data sets in tropical landscapes: cloud and cloud-shadow removal. United States Department of Agriculture— Forest Service: International Institute of Tropical Forestry.
- 74. JSS (2014) ASTER GDEM. Japan Space Systems (J-spacesystems).
- 75. Meyer D (2011) ASTER global digital elevation model version 2–summary of validation result. NASA land processes distributed active archive center and the joint Japan-US ASTER science team.

- McFeeters SK (1996) The use of the normalized difference water index (NDWI) in the delineation of open water features. Int J Remote Sens 17: 1425–1432.
- 77. Breiman L (2001) Random forests. Mach Learn 45: 5–32.
- 78. GoogleInc. (2010) Google Earth (Version). Google.
- Kuenzer C, Dech S (2013) Theoretical background of thermal infrared remote sensing. In: Kuenzer C, Dech S, editors. Thermal infrared remote sensing. Dordrecht: Springer. pp. 1–26.
- Strahler AN (1957) Quantitative analysis of watershed geomorphology. EOS, Trans Am Geophys Union 38: 913–920.
- Di Gregorio A, Jansen LJM (2005) Land cover classification system classification concepts and user manual software version (2). Rome: Food and Agricultural Organization of the United Nations (FAO).
- Karagiannis-Voules DA, Biedermann P, Ekpo UF, Garba A, Langer E, et al. (2015) Spatial and temporal distribution of soil-transmitted helminth infection in sub-Saharan Africa: a systematic review and geostatistical meta-analysis. Lancet Infect Dis 15: 74–84. doi: <u>10.1016/S1473-3099(14)71004-7</u> PMID: <u>25486852</u>
- Jordan P, Webbe G (1993) Epidemiology. In: Jordan P, Webbe G, Sturrock RF, editors. Human schistosomiasis. Wallingford: CAB International. pp. 87–152.
- 84. Schur N, Hürlimann E, Garba A, Traoré MS, Ndir O, et al. (2011) Geostatistical model-based estimates of schistosomiasis prevalence among individuals aged ≤20 years in West Africa. PLoS Negl Trop Dis 5: e1194. doi: 10.1371/journal.pntd.0001194 PMID: 21695107
- Albertson ML, Simons DB (1964) Fluid Mechanics. In: Chow VT, editor. Handbook of applied hydrology: a compendium of water-resources technology. New York, USA: McGraw-Hill. pp. 7–1–7–49.
- Scorza JV, Silva J, Gonzalez L, Machado R (1961) Stream velocity as a gradient in Australorbis glabratus (Say, 1818). Z Tropenmed Parasitol 12: 191–196. PMID: <u>13749320</u>
- Beck-Wörner C, Raso G, Vounatsou P, N'Goran EK, Rigo G, et al. (2007) Bayesian spatial risk prediction of *Schistosoma mansoni* infection in western Côte d'Ivoire using a remotely-sensed digital elevation model. Am J Trop Med Hyg 76: 956–963. PMID: <u>17488922</u>
- Sturrock RF (1993) The intermediate hosts and host-parasite relationships. In: Jordan P, Webbe G, Sturrock RF, editors. Human schistosomiasis. Wallingford: CAB International. pp. 33–85.
- **89.** Sturrock RF (1993) The parasites and their life cycle. In: Jordan P, Webbe G, Sturrock RF, editors. Human schistosomiasis. Wallingford: CAB International. pp. 1–32.
- WHO (1957) Study group on the ecology of intermediate snail hosts of bilharziasis. WHO Tech Rep Ser 120: 1–40.
- Belton S, Stewart TS (2002) Multiple criteria decision analysis: an integrated approach. Massachusetts: Kluwer Academic Publishers.
- 92. Malone JB, Yilma JM, McCarroll JC, Erko B, Mukaratirwa S, et al. (2001) Satellite climatology and the environmental risk of *Schistosoma mansoni* in Ethiopia and East Africa. Acta Trop 79: 59–72. PMID: <u>11378142</u>
- Kabatereine NB, Brooker S, Tukahebwa EM, Kazibwe F, Onapa AW (2004) Epidemiology and geography of Schistosoma mansoni in Uganda: implications for planning control. Trop Med Int Health 9: 372–380. PMID: 14996367
- 94. Boslaugh S, Watters PA (2008) Statistics in a nutshell; Treseler M, editor: O'Reilly Media, Inc.
- **95.** Fisher J, Mustard JF (2004) High spatial resolution sea surface climatology from Landsat thermal infrared data. Remote Sens Environ 90: 293–307.
- Torgersen CE, Faux RN, McIntosh BA, Poage NJ, Norton DJ (2001) Airborne thermal remote sensing for water temperature assessment in rivers and streams. Remote Sens Environ 76: 386–398.
- William DJ, Rybicki NB, Lombana AV, O'Brien TM, Gomez RB (2003) Preliminary investigation of submerged aquatic vegetation mapping using hyperspectral remote sensing. Environ Monit Assess 81: 383–392. PMID: 12620030
- Vahtmäe E, Kutser T, Martin G, Kotta J (2006) Feasibility of hyperspectral remote sensing for mapping benthic macroalgal cover in turbid coastal waters—a Baltic Sea case study. Remote Sens Environ 101: 342–351.
- Marcus WA, Fonstad MA (2008) Optical remote mapping of rivers at sub-meter resolutions and watershed extents. Earth Surf Process Landf 33: 4–24.
- Kiel B, Alsdorf D, LeFavour G (2006) Capability of SRTM C- and X-band DEM data to measure water elevations in Ohio and the Amazon. Photogramm Eng Remote Sens 72: 313–320.

- 101. Clennon JA, King CH, Muchiri EM, Kitron U (2007) Hydrological modelling of snail dispersal patterns in Msambweni, Kenya and potential resurgence of *Schistosoma haematobium* transmission. Parasitology 134: 683–693. PMID: <u>17156580</u>
- 102. King CH (2010) Parasites and poverty: the case of schistosomiasis. Acta Trop 113: 95–104. doi: <u>10.</u> <u>1016/j.actatropica.2009.11.012</u> PMID: <u>19962954</u>
- Butterworth AE (1993) Immunology of schistosomiasis. In: Jordan P, Webbe G, Sturrock RF, editors. Human schistosomiasis. Wallingford: CAB International. pp. 331–366.
- Bundy DAP, Blumenthal U (1990) Human behaviour and the epidemiology of helminth infection. In: Barnard C, Behnke JM, editors. Parasitism and host behaviour. London: Taylor and Francis. pp. 264–289.
- **105.** Schmidlin T, Hürlimann E, Siluée KD, Yapi RB, Houngbedji C, et al. (2013) Effects of hygiene and defection behavior on helminths and intestinal protozoa infections in Taabo, Côte d'Ivoire. PLoS One 8.
- 106. Stothard JR, Mgeni AF, Khamis S, Seto E, Ramsan M, et al. (2002) New insights into the transmission biology of urinary schistosomiasis in Zanzibar. Trans R Soc Trop Med Hyg 96: 470–475. PMID: 12474469